Author's response to reviews

Title: Correlating pharmaceutical data with a national health survey as a proxy for estimating rural population health

Authors:

Ronald E Cossman (rcossman@ssrc.msstate.edu)
Jeralynn S Cossman (Lynne.Cossman@ssrc.msstate.edu)
Wesley L James (Wes.James@memphis.edu)
Troy Blanchard (troy@lsu.edu)
Richard K Thomas (richardkthomas@comcast.net)
Louis G Pol (lpol@mail.unomaha.edu)
Arthur G Cosby (arthur.cosby@ssrc.msstate.edu)

Version: 7 Date: 1 September 2010

Author's response to reviews:

Dear Editor:

Thank you for your careful edit on our manuscript. We have made all the changes you requested. The equation on page 9 was shortened. All authors have reviewed the manuscript and approved it; a statement to that effect was added as well. The references were all checked and every link is a functioning link now. The citation for Table 2 was added, the figure titles were removed from the outlined bozes and we carefully checked for typographical errors.

We look forward to seeing the final product.

Lynne Cossman
Professor of Sociology
Mississippi State University
Lynne.Cossman@msstate.edu